Author's response to reviews

Title: Expertise About Herbs and Dietary Supplements among Diverse Health Professionals

Authors:
Kathi J Kemper (kkemper@wfubmc.edu)
Paula Gardiner (pgardine@massmed.org)
Jessica Gobble (gobble@wfubmc.edu)
Charles Woods (cwoods@wfubmc.edu)

Version: 2 Date: 28 March 2006

Author's response to reviews:

1. The abstract has an expanded background section. We have now used complete sentences throughout the abstract.

2. We have included a competing interests section (none).

3. We have included an Authors' contribution section.